Author’s response to reviews

Title: The use of non-contrast-enhanced MRI to evaluate serial changes in endoleaks after aortic stenting: a case report

Authors:

Yu-Li Lee (a0929103629@gmail.com)
Yao-Kuang Huang (yaokuang@gmail.com)
Li-Sheng Hsu (gv107392@gmail.com)
Pang-Yen Chen (chienwei33@msn.com)
Chien-Wei Chen (chienwei33@gmail.com)

Version: 2 Date: 10 Sep 2019

Author’s response to reviews:

Editor Comments:

1. Please include an abstract after your title page. Your abstract should include the following headings: Background, Case Presentation, Conclusions, and should be no longer than 350 words. Please also include a keywords section, which should three to ten keywords.

Background: the context and purpose of the study
Case Presentation: a brief summary of the case
Conclusions: brief summary and potential implications

Please ensure you apply this format to the abstract in the main text and the submission system also.
We have revised our abstract in the manuscript according to this guideline, and the total number of words is 108.

2. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. E.g. Li-Sheng Hsu would be L-SH.

We have revised the author’s names by their full initials.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

This study was supported by the Chang Gung Memorial Hospital, Taiwan (Contract Nos. CMRP6H0651), including the research projects and instruments. And also the funder is the corresponding author, Chien-Wei Chen, MD, who participated in research design and the manuscript writing.

4. Please remove the response letter, IRB and consent forms, cover letter from the file inventory as it is no longer needed at this stage of the editorial process.

We have removed the response letter, IRB and consent forms and the cover letter.

5. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Thank you for your kind reminders, and we have followed these guidelines carefully.